

Response to the Review's comment

Specific Comments to Authors

It is explained very well with very clear expressions and data that it is almost impossible for such a malposition to occur. I congratulate the authors. It was a clear, understandable, common sense and guiding article. I believe that it should be published specifically as a response to a previously published article.

Thank you very much.